Author's response to reviews

Title: Effect of pathology type and severity on the distribution of MRI signal intensities within the degenerated nucleus pulposus: application to idiopathic scoliosis and spondylolisthesis.

Authors:

Delphine Périé (delphine.perie@polymtl.ca)
Daniel Curnier (daniel.curnier@umontreal.ca)

Version: 5 Date: 17 August 2010

Author's response to reviews:

I am not sure I understand the revision requested.
My references are cited in the text using consecutive numbers in square brackets, not in superscript numbers.
What do you mean by "references are cited out of order"?
Sincerely
Delphine